Author’s response to reviews

Title: Gm40600 suppressed SP 2/0 isograft tumor by reducing Blimp1 and Xbp1 proteins

Authors:

Ying Fang (248511500@qq.com)

Ruonan Xu (1105209894@qq.com)

Bing Zhai (Zhaib80@163.com)

Chunmei Hou (Chunmeihou@yeah.net)

Ning Ma (britty1314@163.com)

Liang Wang (WL1390593786@163.com)

Gencheng Han (genchenghan@yahoo.com)

Zhenyu Jiang (jlujzy@yahoo.com.cn)

Renxi Wang (wang_renxi@hotmail.com)

Version: 1 Date: 11 Oct 2018

Author’s response to reviews:

Dear Editor Rice,

Here is our manuscript entitled " Gm40600 suppressed SP 2/0 xenograft tumor by reducing Blimp1 and Xbp1 proteins" (Ms. No.: BCAN-D-18-02459).

Thank you very much for your reviewed opportunity of the paper. We are appreciated your experts’ suggestions and comments.

According to your suggestions and comments, now we have edited the entire text according to Editor’s suggestions. Our manuscript has been revised according to your style.

Now, we have completely revised the paper and are sending our revised paper to you. We are looking forward to hearing from you.

Thank you very much for your work!

The following is our response to the issues raised by Editor, point by point.
Editor's comments:

1 - Any manuscript submitted to a BioMed Central journal must be original and the manuscript, or substantial parts of it, must not be under consideration by any other journal. BioMed Central is a member of CrossCheck’s plagiarism detection initiative and takes seriously all cases of publication misconduct. We note that the current submission contains textual overlap with other previously published works. Please re-write these sentences and phrases in your own words to ensure no overlap. If there is overlap in the Methods section, please ensure that you summarize the methods and cite the source.

There is 28% overlap with:

BC094916 suppressed SP 2/0 xenograft tumor by down-regulating Creb1 and Bcl2 transcription


There is an overall 46% overlap with other publications.

Authors: Thanks for pointing out these questions. According to your suggestions, we have completely re-write the text.

2 - Please rename the heading Material and Methods to “Methods”

Authors: Thanks for pointing out this question. According to your suggestion, the heading “Material and Methods” have been renamed to “Methods”

3 - In the Funding section of the Declarations please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. If no specific funding was received for this study, please clearly indicate this in the Funding section.

Authors: Thanks for pointing out this question. According to your suggestion, we have added the information “The funding bodies had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.” In the Funding section.